# LIMNOLOGY and OCEANOGRAPHY LETTERS



Open Access

Limnology and Oceanography Letters 2025 © 2025 The Author(s). Limnology and Oceanography Letters published by Wiley Periodicals LLC on behalf of Association for the Sciences of Limnology and Oceanography. doi: 10.1002/lol2.70001

# LETTER

# Anthropogenic nutrient inputs affect productivity–biodiversity relationships in marine tintinnid assemblages

Meiping Feng <sup>(D)</sup>,<sup>1\*</sup> Jichen Qiu,<sup>2</sup> Wenhua Bian,<sup>3,4</sup> Yunfan Zhang <sup>(D)</sup>,<sup>1</sup> Siting Hu,<sup>1</sup> Yijia Yang,<sup>1</sup> Kailin Liu,<sup>5</sup> Chaofeng Wang,<sup>6</sup> Wuchang Zhang,<sup>3</sup> Xiaoxia Sun,<sup>4</sup> Na Li,<sup>1</sup> Jun Lin,<sup>1</sup> Shujin Guo,<sup>4</sup> Bingzhang Chen<sup>7\*</sup>

<sup>1</sup>College of Oceanography and Ecological Science, Shanghai Ocean University, Shanghai, China; <sup>2</sup>State Key Laboratory of Water Environment Simulation, School of Environment, Beijing Normal University, Beijing, China; <sup>3</sup>Key Laboratory of Marine Ecology and Environmental Sciences, Institute of Oceanology, Chinese Academy of Sciences, Qingdao, China; <sup>4</sup>Jiaozhou Bay Marine Ecosystem Research Station, Institute of Oceanology, Chinese Academy of Sciences, Qingdao, China; <sup>5</sup>College of the Environment & Ecology, Xiamen University, Xiamen, China; <sup>6</sup>State Key Laboratory of Mariculture Breeding, Key Laboratory of Marine Biotechnology of Fujian Province, Institute of Oceanology, College of Marine Sciences, Fujian Agriculture and Forestry University, Fuzhou, China; <sup>7</sup>Department of Mathematics and Statistics, University of Strathclyde, Glasgow, UK

# Scientific Significance Statement

The productivity-biodiversity relationship (PBR) is highly controversial, yet insights into marine microzooplankton under varying anthropogenic nutrient inputs have been limited. We verified that PBRs varied across contrasting levels of anthropogenic nutrient input, and high productivity promoted greater diversity under low to medium levels than under high levels of anthropogenic nutrient input. Compared to conventional species richness, functional and phylogenetic diversity respond more quickly to nutrient changes and differently to productivity in both long-term and monthly trends, which deliver more information on PBRs and the underlying ecological processes. We also empirically addressed the previous concern regarding methodological inconsistencies in the meta-analysis of PBRs, providing one of the first consistent observational datasets on PBRs in marine unicellular secondary producers. Our study contributes to a comprehensive understanding of PBRs in microzooplankton under contrasting anthropogenic nutrient input conditions.

# **Abstract**

Although the productivity-biodiversity relationship (PBR) has been a hot topic, few studies have considered how anthropogenic pressures affect PBRs in marine microzooplankton. Here, we provide the first insights into PBRs in tintinnid assemblages using 18-yr data from Jiaozhou Bay, a typical coastal bay in the Yellow Sea. We hypothesized and verified that PBRs vary across contrasting anthropogenic nutrient inputs and that functional and phylogenetic diversity would deliver more information than conventional species richness. High productivity promotes more diversity under low to medium rather than high anthropogenic nutrient inputs. Compared

#### Associate editor: Raphael Kudela

Data Availability Statement: Data used in the analysis are available in the Dryad repository at https://doi.org/10.5061/dryad.k98sf7mdw (Reviewer URL).

<sup>\*</sup>Correspondence: fmeiping@126.com and bingzhang.chen@strath.ac.uk

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

23782242, 0, Downloaded from https://aslopubs.onlinelibrary.wiley.com/doi/10.

. 1002/IoI2.70001 by NHS Education for Scotland NES, Edinburgh Central Office, Wiley

Online Library on [21/02/2025]. See the Terms

and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

to species richness, functional and phylogenetic diversity reveal more PBR patterns and respond more quickly in response to varying anthropogenic inputs. A concave+ PBR is revealed for functional diversity in the ecozone with highly active water exchange. Our study contributes to the understanding of PBR in marine unicellular secondary producers and their responses to anthropogenic nutrient inputs in coastal ecosystems.

The productivity-biodiversity relationship (PBR) has been highly controversial over the past few decades (Whittaker 2010; Adler et al. 2011; Willig 2011; Grace et al. 2016; Brun et al. 2019). Five general patterns of PBRs have been distilled, including increasing (positive), humped (unimodal, concave-), decreasing (negative), U-shaped (concave+), and no relationship; however, these patterns, along with their causal processes, face critique (Gillman and Wright 2006; Whittaker 2010). Meta-analyses have been employed to discern the mechanisms underlying these patterns (Mittelbach et al. 2001; Gillman and Wright 2006), but they have raised new problems such as methodological inconsistencies, generating highly divergent outcomes, inconsistent classification of data sets, and mega-mistakes from one meta-analysis to the next (Whittaker 2010).

Biodiversity encompasses multiple attributes such as taxonomic, phylogenetic, and functional diversities, which may have contrasting effects on productivity (Naeem et al. 2016; Le Bagousse-Pinguet et al. 2019). Conventional biodiversity indices (e.g., species richness and Shannon index) without considering species similarity may lead to suspicion about their ecological significance, as species differ in their functions and ecological strategies (Stirling 2007; Leinster and Cobbold 2012). The niches and competitive abilities of species within a community are often reflected in functional and phylogenetic differences, suggesting that functional and phylogenetic diversity may provide a deeper understanding of PBRs (Mayfield and Levine 2010; Brun et al. 2019). These dimensions of biodiversity can reveal underlying ecological processes, such as environmental filtering, competitive exclusion, and evolutionary context (Mayfield and Levine 2010; Brun et al. 2019), as well as responses to nutrient availability and anthropogenic inputs (Prowe et al. 2012; Gamfeldt et al. 2015; Lehtinen et al. 2017), and further support the biodiversity-ecosystem multifunctionality relationship (Flynn et al. 2011; Naeem et al. 2016; Le Bagousse-Pinguet et al. 2019).

Most previous PBR studies have focused on terrestrial macroorganisms, especially multicellular plants, that is, primary producers (e.g., Adler et al. 2011; Willig 2011; Grace et al. 2016; Brun et al. 2019), whereas data on vast marine unicellular plankton remain limited (e.g., Irigoien et al. 2004; Witman et al. 2008). For marine phytoplankton, there is less data available, but a few large-scale studies also suggest that humped PBRs, with maximum diversity occurring at intermediate productivity levels (Li 2002; Irigoien et al. 2004; Smith 2007; Vallina et al. 2014), exhibit significant spatial and temporal variation and relate to nutrient availability (Vadrucci et al. 2003; Lehtinen et al. 2017) or in some cases, show no relationship (Cermeño et al. 2013).

In contrast, current knowledge regarding the PBRs for microzooplankton (20–200  $\mu$ m in size) remains limited. As the main consumers of marine primary production, microzooplankton can grow and reproduce rapidly, may respond more quickly to environmental changes, and are better coupled to ecological processes than slow-responding metazoans (Calbet and Landry 2004). However, PBRs of microzooplankton remain hitherto unexplored and highly fragmentary and are mainly based on meta-analyses and sensitive to heterogeneous datasets (Irigoien et al. 2004; Ibarbalz et al. 2019). As a model of plankton ecology, tintinnids are an ideal tool for microzooplankton to explore PBRs with functional and phylogenetic diversity, attributed to their loricae, which are of taxonomic significance and related to ecological characteristics and functional responses (Dolan et al. 2013). Tintinnids have been increasingly used as bioindicators of anthropogenic impacts and environmental changes in plankton ecosystems (e.g., Dolan et al. 2016; Al-Yamani et al. 2019; Wang et al. 2024).

Our study provides one of the first empirical assessments of how PBRs incorporating multidimensional biodiversity respond to contrasting anthropogenic nutrient inputs (e.g., dissolved inorganic nitrogen and phosphate) in tintinnids over an 18-yr cycle. In addition to conventional species richness, we employed functional and phylogenetic diversity indices following Brun et al. (2019). Compared to the terrestrial multicellular primary producers studied by Brun et al. (2019), our work offers a new perspective on PBR patterns for marine unicellular secondary producers (microzooplankton) and examines their responses to anthropogenic nutrient inputs. To obtain convincing evidence, we used consistent, continuous observational sampling in a semi-enclosed coastal bay. Our study contributes to a comprehensive understanding of PBRs in marine microzooplankton, particularly in the context of varying anthropogenic nutrient inputs to temperate coastal ecosystems.

#### **Methods**

#### Study site

We examined tintinnid assemblages in 1280 continuous samples from Jiaozhou Bay Marine Ecosystem Research Station (https://jzw.qdio.cas.cn/) at 22 sites in Jiaozhou Bay (JZB, 120°04'E–120°23'E, 35°38'N–36°18'N, Fig. 1) from May 2003 to December 2020. Jiaozhou Bay is a semi-enclosed coastal bay with a narrow channel (about 3 km) connecting to the

Yellow Sea (Liu et al. 2004). With heavy anthropogenic impacts and frequent red tides. JZB has served as an ideal model system to explore the effects of ongoing climate changes and anthropogenic inputs on marine ecosystems (e.g., Feng et al. 2015; Guo et al. 2021; Wang et al. 2023). Anthropogenic nutrient inputs contribute more nutrients to JZB, primarily by rivers, than atmospheric deposition, accounting for most dissolved inorganic nitrogen and phosphate (Liu et al. 2005; Xu et al. 2020). The JZB witnessed the sharp decline of anthropogenic nutrient inputs in 2008 (Xu et al. 2020), with the decreasing phytoplankton, increasing zooplankton, and miniaturization of copepods during 2002–2016 (Wang et al. 2023). We split the sampling sites according to four levels of anthropogenic nutrient inputs based on the division of Zhao et al. (2020) from lowest AP1 to highest AP4 (Fig. 1, Table 1). Anthropogenic nutrient inputs were lowest in AP1 in the open waters outside the bay farthest from the rivers. The hydrodynamic condition was unique for AP2 in the bay mouth with highly active water exchange. Anthropogenic nutrient inputs were relatively high in AP3 surrounded by many seasonal rivers (medium level) and peaked in AP4 in the bay with the primary region for shellfish and algae farming and the largest river, Dagu River, flowing into.

#### Data collections and calculation

The sampling, identification, and enumeration of tintinnids were conducted and summarized in Part 1 in the Supporting Information. The counting effort ranged from 1 to 330 individuals per sample. We used biomass as a proxy for plankton productivity, consistent with most empirical studies (e.g., Groner and Novoplansky 2003; Vadrucci et al. 2003; Irigoien et al. 2004; Smith 2007; Vallina et al. 2014). Tintinnid biomass was calculated using the equation: C (pg) = 444.5 + 0.053 LV (Lorica Volume, Verity and Langdon 1984)  $\mu$ g carbon per L (Table S1). Functional traits and phylogenetic information of tintinnids were summarized (Tables S1, S2) based on microscope observations in this study with reference to the literature (Zhang et al. 2012; Dolan et al. 2013; Santoferrara and Mcmanus 2020; Agatha and Bartel 2022), encompassing vulnerability traits, life history traits, and habitat use.

For diversity, we focused on the common indices, including species richness and quadratic entropies of functional and phylogenetic diversity. To calculate functional and phylogenetic diversity, we employed the workflow of biodiversity space following Brun et al. (2019), by successively defining the scale of species similarity ( $\delta$ ), type of similarity ( $\alpha$ ), and species dominance effect (q) using functional traits and phylogenetic information (Tables S1, S2). We derived LC<sub>funct</sub> ( $\delta = 1$ ,  $\alpha = 1$ , q = 2) and LC<sub>phylo</sub> ( $\delta = 1$ ,  $\alpha = 0$ , q = 2) for functional and phylogenetic diversity, respectively, following Leinster and Cobbold (2012) and Chen and Grinfeld (2024), with detailed procedures in Part 1 in the Supporting Information. All steps were repeated for different groups of communities under four levels of anthropogenic nutrient inputs.



**Fig. 1.** Study area considering contrasting anthropogenic nutrient inputs in the coastal bay, Jiaozhou Bay (JZB) in the Yellow Sea in northern China from 2003 to 2020. Jiaozhou Bay is a semi-enclosed water body (390 km<sup>2</sup>, averaged depth of 7 m) with a narrow channel (about 3 km) connecting to the Yellow Sea (Liu et al. 2004). Anthropogenic nutrient inputs were lowest in AP1 in the open waters outside the bay farthest from the rivers, increased from AP1 to AP4, and were heaviest in AP4 in the bay where the largest river, Dagu River, flows into (Zhao et al. 2020). The hydrodynamic condition was unique for AP2 in the bay mouth with highly active water exchange. Details are provided in Table 1.

Region Chá AP1 Offsh the rep op				Productivity (I	og-transformed						
Region Cha AP1 Offshu the rep op			Sampling -	bior	nass)	Ŧ	lichness		LC <sub>funct</sub>	Ϋ́	hylo
AP1 Offshr the rep op	aracteristics	Sites	periods n	Mean	<i>p</i> value	Mean	<i>p</i> value	Mean	<i>p</i> value	Mean	<i>p</i> value
adj	ore area of : Yellow Sea, rresenting en sea and acent waters.	JZB10, JZB12, JZB13	2003-2020 449	$-0.73 \pm 0.71$ (-2.83-1.53)	AP1-AP2 0.001, AP1-AP3 0.004, AP1-AP4 0.305	6.46 ± 3.4 (1−21)	AP1-AP2 0.937, AP1-AP3 0.533, AP1-AP4 0.005	2 ± 0.79 (1–6.26)	AP1-AP2 0.002, AP1-AP3 0.000, AP1-AP4 0.000	1.24 ± 0.14 <i>i</i> (1−1.69)	P1-AP2 0.159, AP1-AP3 0.021, AP1-AP4 0.002
AP2 Deep chi mc free ex ex ex bio bio	-water annel at bay buth, with quent water thange and nificant sical- logical inges.	A2, A4, B3, B4, C2, JZB07, JZB08, JZB09	2003-2005 181	-0.46 ± 0.73 (-2.42-0.99)	AP2-AP3 0.500, AP2-AP4 0.000	$6.28 \pm 3.04$ (1-17)	AP2-AP3 0.979, AP2-AP4 0.049	1.77 ± 0.64 (1-4.53)	AP2.AP3 0.996, AP2-AP4 0.188	1.21 ± 0.13 / (1-1.6)	P2-AP3 1.000, AP2-AP4 0.269
AP3 Inner sur ma mc rive	bay, rounded by ny rivers, stly seasonal rs.	B6, D2, D4, JZB02, JZB03, JZB05, JZB06, JZB14	2003-2020 563	$-0.56 \pm 0.82$ (-3.03-1.47)	AP3-AP4 0.002	$6.17 \pm 3.54$ (1-21)	AP3-AP4 0.043	$1.79 \pm 0.68$ (1-4.63)	AP3-AP4 0.071	$1.21 \pm 0.15$ , (1-1.69)	AP3-AP4 0.168
AP4 Inner infl in J (Di (Di ser for alg	bay uenced by i largest river iaozhou Bay igu River), ving as a mary region shelffish and ae farming.	JZB01, JZB04, JZB11	2003-2005 87	-0.89 ± 1.01 (-3.05-1.19)		5.14 ± 3.13 (1-12)		1.59 ± 0.55 (1-2.92)		1.18 ± 0.15 (1–1.63)	

Table 1. Basic information for the samplings under the four levels of anthropogenic nutrient inputs in the coastal bay, Jiaozhou Bay (JZB) in the Yellow Sea in

nutrient input levels were sourced from Zhao et al. (2020), with consideration of Xu et al. (2020).

23782222, 0, Downloaded for https://solopto.onlinelibrary.wiley.com/doi/1.1022/02.70001 by NHS Education for Soduad NES, Edinburgh Central Office, Wiely Online Library on [21/02/02]. See the Terms and Conditions (https://solaideibrary.wiley.com/term-and-conditions) on Wiely Online Library for rules of use; OA articles are governed by the applicable Centre Commons License

Functional traits and phylogenetic information of tintinnids were summarized (Tables S1, S2) based on microscope observations conducted in this study, with reference to the literature (Zhang et al. 2012; Dolan et al. 2013; Santoferrara and Mcmanus 2020; Agatha and Bartel 2022). These summaries encompassed vulnerability traits, life history traits, and habitat use.

#### Statistical analyses

We first assessed the significant long-term trends for productivity (biomass) and biodiversity with the Mann–Kendall test using the R package "Kendall" (Frenken et al. 2023). Then we explored the PBRs using generalized additive models (GAMs, *see* details in Part 1 in the Supporting Information) in R after checking for the outliers (Adler et al. 2011). MANOVA was applied to test the effects of anthropogenic nutrient inputs on productivity and biodiversity. Akaike's Information Criterion (AIC) and explained variances were deployed to evaluate the models, excluding variables with high concurvity (the GAM equivalent of collinearity). No significant spatial autocorrelation was found after including latitude and longitude as extra predictors in the models, following Le Bagousse-Pinguet et al. (2019).

We used REML to estimate model coefficients. The PBRs were then classified into different classes based on the goodness of fit ( $R^2$ ) and curve shape criteria (Table S5) according to Brun et al. (2019). Those curves with  $R^2$  <0.15 were labeled as non-significant as per Brun et al. (2019).

# Results

#### General variations of productivity and biodiversity

The productivity and three focal diversity measures of the tintinnid assemblages varied significantly across the four levels of anthropogenic nutrient inputs (MANOVA, p < 0.01). Tintinnids under the highest anthropogenic nutrient input (AP4) showed relatively low productivity and diversity compared to those under medium levels (AP2 and AP3), with significant differences in productivity (p < 0.01) and species richness (p < 0.05). However, no significant differences were observed in the functional or phylogenetic diversity (Table 1; p > 0.05). In contrast, at the lowest anthropogenic input level (AP1), tintinnids exhibited significantly lower productivity than at medium levels (AP2 and AP3, p < 0.01), but higher values at all three biodiversity levels than at other levels. Significant differences were observed in terms of functional diversity between AP1 and each of the other three levels (p < 0.001) but only between AP1 and AP4 in terms of species richness (*p* < 0.01).

In addition, the GAM results revealed that temporal variability (both long-term and monthly) affected the productivity and diversity of tintinnid assemblages (Table S3). Similar to productivity, species richness and phylogenetic diversity showed a small peak in 2008, followed by a general decrease, whereas functional diversity showed a general increase (Fig. S1).

#### Effect of anthropogenic nutrient inputs on the PBRs

As for the PBRs, the GAM results (Table S4) showed that levels of anthropogenic nutrient inputs and the interaction effects of productivity, month, and year were all significant predictors of species richness ( $R^2 = 0.64$ ; deviance explained 65%), functional diversity ( $R^2 = 0.22$ ; deviance explained 23%), and functional diversity ( $R^2 = 0.34$ ; deviance explained 36%). Similar trends in the interaction between productivity and month were revealed for the three biodiversity indices, showing greater effects on diversity in August, with the highest productivity. In contrast, functional and phylogenetic diversity responded quickly to the sharp decline in anthropogenic nutrient inputs in 2008, with the former increasing (Fig. 2d) and the latter decreasing (Fig. 2f). Species richness responded more slowly and increased only after 2010 (Fig. 2b).

After keeping the other parameters constant, the relationships between productivity and the three biodiversity indices displayed contrasting patterns across different levels of anthropogenic nutrient input (Fig. 3). For species richness, increasing PBRs prevailed at low to medium levels of anthropogenic nutrient inputs (Fig. 3a–c), whereas a concave– PBR was revealed in AP4, which had the highest anthropogenic nutrient inputs (Fig. 3d). Similarly, under the highest level of anthropogenic nutrient input, concave– PBRs were found for both functional and phylogenetic diversity (Fig. 3h,l). All three diversity indices showed a concave– PBR at the highest level, revealing that high anthropogenic nutrient inputs promoted decreased diversity at high productivity.

Compared to species richness, functional and phylogenetic diversity revealed a new perspective on PBRs under different levels of anthropogenic nutrient inputs. Species richness failed to show evident changes in PBRs (all increasing) between AP1 and AP3; however, both functional and phylogenetic diversity revealed different PBRs. For AP1, under the lowest anthropogenic inputs, both functional and phylogenetic diversity revealed a concave– PBR (Fig. 3e,i), showing decreasing functional and phylogenetic diversity along with increasing species richness at high productivity. Notably, functional diversity revealed a distinct PBR pattern from others, concave + PBRs, in AP2 at the bay mouth with highly active water exchange and medium anthropogenic inputs, showing low functional diversity at intermediate levels of productivity (Fig. 3f).

The interaction between productivity and month significantly affected diversity (Table S4; p < 0.01), with its effects varying across the four levels of anthropogenic nutrient input (Fig. 4). For species richness, a pronounced effect was observed around August, coinciding with high productivity. In contrast, the effects were broader for functional and phylogenetic diversity, particularly for functional diversity at high levels of anthropogenic nutrient input (Fig. 4h) and phylogenetic



**Fig. 2.** Generalized additive model response of diversity to the interactions between productivity and month, year, and levels of anthropogenic nutrient inputs (AP) in Jiaozhou Bay during the studied period. Biomass (log-transformed,  $\mu g/L$ ) was used as a proxy of productivity following Irigoien et al. (2004). Diversity including species richness (**a–b**), functional diversity (LCfunct, **c–d**), and phylogenetic diversity (LCphylo, **e–f**) are compared among months and years. Overlaid curves show the partial effect of productivity (log-transformed biomass) on the diversity in GAM fits with 95% confidence intervals. LCfunct and LCphylo represent functional and phylogenetic diversity using LC biodiversity values at ( $\delta = 1$ ,  $\alpha = 1$ , q = 2) and ( $\delta = 1$ ,  $\alpha = 0$ , q = 2), respectively.

diversity at medium to high levels of anthropogenic nutrient input (Fig. 4k,l).

#### Discussion

Our study firstly verified that PBRs varied across contrasting levels of anthropogenic nutrient input in unicellular microzooplankton assemblages in coastal marine ecosystems. We found that increasing PBRs prevail at low to medium levels of anthropogenic nutrient input for species richness, indicating that high productivity promotes diverse communities with low to medium levels of anthropogenic nutrient input. This corresponds to the niche differentiation and limiting similarity hypotheses, which assume that competitive exclusion favors complementary strategies and promotes assemblages with diverse traits (Pacala and Tilman 1994; Brun et al. 2019).

In contrast, concave– (humped) PBRs prevailed in AP4 for all three diversity indices, indicating that with high anthropogenic nutrient inputs, diversity increased at low productivity and decreased at high productivity. This is consistent with the speculation that humped PBRs in marine zooplankton result from a balance between food limitations at low population levels and selective predation at high population levels (Irigoien et al. 2004; Vallina et al. 2014). According to the most prominent historical hypothesis, biodiversity is low in unproductive environments because of environmental filtering, increases with increasing resource supply until it reaches the coexistence limits at high productivity, and declines because of increased competition and competitive exclusion (Grime 1973; Al-Mufti et al. 1977; Grace et al. 2016; Brun et al. 2019). Productive communities are less diverse with high anthropogenic nutrient inputs, probably because competition can eliminate more distinct and less related taxa (Mayfield and Levine 2010). In addition, this result may provide another possible explanation for the negative PBRs of global marine plankton (Ibarbalz et al. 2019), since they used phylogenetic diversity, rather than taxonomic diversity, which is more related to previously reported functional traits.

We further verified that, in response to varying anthropogenic nutrient inputs, functional and phylogenetic diversity offers more insights into PBRs and underlying ecological

23782242, 0, Downloaded from https

://aslopubs.onlinelibrary.wiley.com/doi/10.1002/lol2.70001 by NHS Education

for Scotland NES, Edinburgh Central Office, Wiley Online Library on [21/02/2025]. See the Terms

and Condition

use; OA articles are

governed by the applicable Creative Commons



**Fig. 3.** Generalized additive model response curves of diversity to productivity of tintinnid assemblages at four levels of anthropogenic nutrient inputs in Jiaozhou Bay during the studied period. Biomass (log-transformed,  $\mu$ g/L) was used as a proxy of productivity following Irigoien et al. (2004). Diversity including species richness (**a–d**), LC<sub>funct</sub> (**e–h**), and LC<sub>phylo</sub> (**i–l**) are compared between communities of different levels of anthropogenic nutrient inputs: low (AP1, a, e, i), medium with active water exchange (AP2), medium (AP3), and high (AP4). Overlaid curves show the partial effect of productivity (log-transformed biomass) on the diversity in GAM fits with 95% confidence intervals. Overlaid curves are colored according to curve type classification: red is concave–; blue is increasing; dark green is Concave+. LC<sub>funct</sub> and LC<sub>phylo</sub> represent functional and phylogenetic diversity using LC biodiversity values at ( $\delta = 1$ ,  $\alpha = 0$ , q = 2), respectively.

processes than conventional species richness, especially at low to medium levels of anthropogenic nutrient inputs. We found more PBR patterns in functional and phylogenetic diversity at low to medium levels of anthropogenic nutrient input, which were not revealed for species richness. Furthermore, functional and phylogenetic diversity responded faster than species richness to the reduction in anthropogenic nutrient loads around 2008. This is consistent with the expected importance of functional and phylogenetic diversity in supporting the biodiversity–ecosystem multifunctionality relationship (Flynn et al. 2011; Le Bagousse-Pinguet et al. 2019).

Notably, the concave+ (U-shaped) PBR in AP2 had not been extensively discussed before the meta-analysis work by Mittelbach et al. (2001), and the interaction of productivitydriven processes with other processes may be a probable explanation (Scheiner and Jones 2002). This also explains why we only found the concave+ PBR in AP2, the ecozone of all the other three levels of anthropogenic nutrient inputs and highly active water exchange. Nevertheless, different scales and statistical methods have been reported to lead to different classifications of concave+ PBRs (Mittelbach et al. 2001), and some have been classified as humped or having no relationship (Mittelbach et al. 2001; Gillman and Wright 2006; Whittaker 2010). The classification and causal processes of PBRs remain highly controversial (Gillman and Wright 2006). Most of the associated hypotheses on PBRs are derived from terrestrial systems and are probably limited when applied to marine systems.



**Fig. 4.** Generalized additive model response of diversity to the interaction between productivity and month at four levels of anthropogenic inputs in Jiaozhou Bay during the studied period. Biomass (log-transformed,  $\mu$ g/L) was used as a proxy of productivity following Irigoien et al. (2004). Diversity including species richness (**a–d**), LC<sub>funct</sub> (**e–h**), and LC<sub>phylo</sub> (**i–I**) are compared between communities of different levels of anthropogenic nutrient inputs: low (AP1, a, e, i), medium with active water exchange (AP2), medium (AP3), and high (AP4). LC<sub>funct</sub> and LC<sub>phylo</sub> represent functional and phylogenetic diversity using LC biodiversity values at ( $\delta = 1$ ,  $\alpha = 1$ , q = 2) and ( $\delta = 1$ ,  $\alpha = 0$ , q = 2), respectively.

Anthropogenic nutrient inputs would affect the PBRs of plankton communities through trophic links, including topdown and bottom-up controls (Vallina et al. 2014). Serving as secondary producers and key links between phytoplankton and copepods, tintinnid communities decreased in productivity, species richness, and phylogenetic diversity, in line with generally decreasing nutrients and phytoplankton, and increasing zooplankton during the investigated period (Wang et al. 2023). However, our study also found increasing functional diversity for tintinnids, especially after 2008; that is, tintinnids were more diverse in ecological functions and less diverse in richness and phylogeny, which might be due to the generally decreasing abundance ratio of large-sized copepods. The competition vulnerability trade-off (top-down control) allows stable predator-mediated coexistence of species within and across functional groups, whereas the growth-affinity trade-off (bottom-up control) allows the non-equilibrium coexistence of functional groups ranging from nutrient specialists to nutrient opportunists, resulting in humped PBRs in phytoplankton (Vallina et al. 2014). The positive effect of nutrient supply on species diversity only occurred in the presence of predators because the two processes do not act in isolation (Worm et al. 2002). Other processes should be considered to explain the potential contrasting responses of PBRs to anthropogenic nutrient inputs, including the

sampling regime, scale of analysis, and processes such as community assembly.

We acknowledge the limitation of the use of tintinnid biomass as a productivity proxy. In PBR studies, productivity is estimated either directly from the organisms' biomass or energy, or indirectly through lower trophic level productivity and surrogates, with the PBR pattern being sensitive to the directness of productivity estimates in animal studies but not in plant studies (Groner and Novoplansky 2003). Herein we used biomass as a productivity proxy, consistent with most empirical studies (e.g., Irigoien et al. 2004; Vallina et al. 2014), but there are limitations to this approach. We stressed biomass (for more contribution in the productivity) rather than rate, among the three aspects of productivity (Clarke et al. 1946), because rate relies on empirical data and temperature and might introduce more bias in the productivity estimation in our long-term samples spanning large temperature ranges. Rate (production rate) is also important for tintinnid productivity, calculated as the product of biomass and empirically specific growth rates (g) for tintinnids (Godhantaraman 2002). However, g was derived from freshwater ciliates, which differ from marine tintinnids in our study in thermal performance and growth rates (Lukić et al. 2022). Furthermore, growth rates differ between tintinnids and ciliates: they increase allometrically with cellular biomass for tintinnids

#### Feng et al.

(Heinbokel 1978; Stoecker et al. 1983; Verity 1985, 1986; Godhantaraman 2002), but not for all planktonic ciliates (Lukić et al. 2022; Weisse 2024). Allometric considerations remain important in planktonic food webs, as direct methods for measuring tintinnid growth in natural assemblages are still limited (Dolan et al. 2013). In summary, this proxy should be used with caution, and additional field data are needed for more robust conclusions on tintinnids and other microzooplankton.

Additionally, meta-analyses have been increasingly emploved as powerful tools for PBR studies, but they have also been criticized for relying on data collected from previous studies using inconsistent methods (Whittaker 2010; Adler et al. 2011). Our study provides one of the first insights into PBRs in marine microzooplankton using continuous data obtained by conducting consistent and continuous observational sampling, which further empirically contributes to addressing methodological inconsistencies and convincing conclusions. Nevertheless, it is important to acknowledge that standardized observational studies have limitations, such as uncontrolled environmental factors, which restrict the analysis to correlations rather than causations and fail to address the paradox: Is diversity the cause or consequence of productivity? (Cardinale et al. 2009). Additionally, in relation to resource use strategies and ecological processes, the functional traits in our study were limited, lacking aspects such as trophic structure and disturbance (Hillebrand and Matthiessen 2009; Grace et al. 2016), which also led to limitations in the statistical analysis. Further exploration and integrative modeling are required to confirm our findings and explore the variation in PBRs under anthropogenic impacts in coastal marine ecosystems.

## **Author Contributions**

Meiping Feng and Bingzhang Chen conceived of the presented idea. Meiping Feng, Wenhua Bian, Wuchang Zhang, Chaofeng Wang, Xiaoxia Sun, and Shujin Guo conducted the field samplings and lab work. Meiping Feng, Wenhua Bian, Jichen Qiu, Yunfan Zhang, Siting Hu, Yijia Yang, Na Li, and Jun Lin compiled the original data. Jichen Qiu, Meiping Feng, Bingzhang Chen, and Kailin Liu performed statistical analysis. Meiping Feng wrote the manuscript with support from Bingzhang Chen. All authors critically reviewed and approved the manuscript.

## Acknowledgments

The data were generated with the help of Jiaozhou Bay National Marine Ecosystem Research Station, Institute of Oceanology, Chinese Academy of Sciences. This work was supported by the National Natural Science Foundation of China (NSFC Nos. 41806160, 32371619, 42206150), Leverhulme Trust rpg-2020-389, and the Science and Technology Development Program of Shanghai Ocean University (A2-2006-21-200319 and A2-2006-20-200207).

#### **Conflicts of Interest**

None declared.

#### References

- Adler, P. B., E. W. Seabloom, and E. T. Borer. 2011. "Productivity Is a Poor Predictor of Plant Species Richness." *Science* 333: 1750–1753. https://doi.org/10.1126/science.1204498.
- Agatha, S., and H. Bartel. 2022. "A Comparative Ultrastructural Study of *Tintinnid loricae* (Alveolata, Ciliophora, Spirotricha) and a Hypothesis on their Evolution." *Journal of Eukaryotic Microbiology* 69: e12877. https://doi.org/10.1111/ jeu.12877.
- Al-Mufti, M. M., C. L. Sydes, S. B. Furness, J. P. Grime, and S. R. Band. 1977. "A Quantitative Analysis of Shoot Phenology and Dominance in Herbaceous Vegetation." *Journal of Ecology* 65: 759. https://doi.org/10.2307/2259378.
- Al-Yamani, F., R. Madhusoodhanan, V. Skryabin, and T. Al-Said. 2019. "The Response of Microzooplankton (Tintinnid) Community to Salinity Related Environmental Changes in a Hypersaline Marine System in the Northwestern Arabian Gulf." *Deep Sea Research Part II: Topical Studies in Oceanography* 166: 151–170. https://doi.org/10.1016/j. dsr2.2019.02.005.
- Brun, P., N. E. Zimmermann, C. H. Graham, et al. 2019. "The Productivity–Biodiversity Relationship Varies Across Diversity Dimensions." *Nature Communications* 10: 5691. https:// doi.org/10.1038/s41467-019-13,678-1.
- Calbet, A., and M. R. Landry. 2004. "Phytoplankton Growth, Microzooplankton Grazing, and Carbon Cycling in Marine Systems." *Limnology and Oceanography* 49: 51–57. https:// doi.org/10.4319/lo.2004.49.1.0051.
- Cardinale, B. J., D. M. Bennett, C. E. Nelson, and K. Gross. 2009. "Does Productivity Drive Diversity or Vice Versa? A Test of the Multivariate Productivity–Diversity Hypothesis in Streams." *Ecology* 90: 1227–1241. https://doi.org/10.189 0/08-1038.1.
- Cermeño, P., T. Rodríguez-Ramos, M. Dornelas, et al. 2013. "Species Richness in Marine Phytoplankton Communities Is Not Correlated to Ecosystem Productivity." *Marine Ecology Progress Series* 488: 1–9. https://doi.org/10.3354/mep s10443.
- Chen, B., and M. Grinfeld. 2024. "Decomposition of the Leinster-Cobbold Diversity Index." *Ecology and Evolution* 14: e10952. https://doi.org/10.1002/ece3.10952.
- Clarke, G. L., W. T. Edmondson, and W. E. Ricker. 1946. "Dynamics of Production in a Marine Area." *Ecological Monographs* 16: 321–337. https://doi.org/10.2307/1961639.
- Dolan, J. R., D. J. S. Montagnes, S. Agatha, W. D. Coats, and D. K. Stoecker, eds. 2013. *The Biology and Ecology of*

*Tintinnid Ciliates: Models for Marine Plankton*. Chichester: Wiley.

- Dolan, J. R., E. J. Yang, S.-H. Kang, and T. S. Rhee. 2016. "Declines in both Redundant and Trace Species Characterize the Latitudinal Diversity Gradient in Tintinnid Ciliates." *ISME Journal* 10: 2174–2183. https://doi.org/10.1038/ ismej.2016.19.
- Feng, M., W. Zhang, W. Wang, G. Zhang, T. Xiao, and H. Xu. 2015. "Can Tintinnids be Used for Discriminating Water Quality Status in Marine Ecosystems?" *Marine Pollution Bulletin* 101: 549–555. https://doi.org/10.1016/j.marpolbul. 2015.10.059.
- Flynn, D. F. B., N. Mirotchnick, M. Jain, M. I. Palmer, and S. Naeem. 2011. "Functional and Phylogenetic Diversity as Predictors of Biodiversity–Ecosystem–Function Relationships." *Ecology* 92: 1573–1581. https://doi.org/10.1890/10-1245.1.
- Frenken, T., K. M. Brandenburg, and D. B. Van De Waal. 2023. "Long-term nutrient load reductions and increasing lake TN : TP stoichiometry decrease phytoplankton biomass and diversity in a large shallow lake." *Limnology and Oceanography* 68, no. 10: lno.12428. https://doi.org/10. 1002/lno.12428.
- Gamfeldt, L., J. S. Lefcheck, J. E. K. Byrnes, B. J. Cardinale, J. E. Duffy, and J. N. Griffin. 2015. "Marine Biodiversity and Ecosystem Functioning: What's Known and What's Next?" *Oikos* 124: 252–265. https://doi.org/10.1111/oik.01549.
- Gillman, L. N., and S. D. Wright. 2006. "The Influence of Productivity on the Species Richness of Plants: A Critical Assessment." *Ecology* 87: 1234–1243. https://doi.org/10.189 0/0012-9658(2006)87[1234:TIOPOT]2.0.CO;2.
- Godhantaraman, N. 2002. "Seasonal Variations in Species Composition, Abundance, Biomass and Estimated Production Rates of Tintinnids at Tropical Estuarine and Mangrove Waters, Parangipettai, Southeast Coast of India." *Journal of Marine Systems* 36: 161–171. https://doi.org/10. 1016/S0924-7963(02)00185-9.
- Grace, J. B., T. M. Anderson, and E. W. Seabloom. 2016. "Integrative Modelling Reveals Mechanisms Linking Productivity and Plant Species Richness." *Nature* 529: 390–393. https:// doi.org/10.1038/nature16524.
- Grime, J. P. 1973. "Competitive Exclusion in Herbaceous Vegetation." *Nature* 242: 344–347. https://doi.org/10.1038/242 344a0.
- Groner, E., and A. Novoplansky. 2003. "Reconsidering Diversity–Productivity Relationships: Directness of Productivity Estimates Matters." *Ecology Letters* 6: 695–699. https://doi.org/10.1046/j.1461-0248.2003.00488.x.
- Guo, S., Z. Zhao, J. Liang, J. Du, and X. Sun. 2021. "Carbon Biomass, Carbon-to-Chlorophyll a Ratio and the Growth Rate of Phytoplankton in Jiaozhou Bay, China." *Journal of Oceanology and Limnology* 39: 1328–1342. https://doi.org/ 10.1007/s00343-020-0234-z.
- Heinbokel, J. F. 1978. "Studies on the Functional Role of Tintinnids in the Southern California Bight. I. Grazing and

Growth Rates in Laboratory Cultures." *Marine Biology* 47: 177–189. https://doi.org/10.1007/BF00395638.

- Hillebrand, H., and B. Matthiessen. 2009. "Biodiversity in a Complex World: Consolidation and Progress in Functional Biodiversity Research." *Ecology Letters* 12: 1405–1419. https://doi.org/10.1111/j.1461-0248.2009.01388.x.
- Ibarbalz, F. M., N. Henry, and M. C. Brandão. 2019. "Global Trends in Marine Plankton Diversity Across Kingdoms of Life." *Cell* 179: 1084–1097. https://doi.org/10.1016/j.cell. 2019.10.008.
- Irigoien, X., J. Huisman, and R. P. Harris. 2004. "Global Biodiversity Patterns of Marine Phytoplankton and Zooplankton." *Nature* 429: 863–867. https://doi.org/10.1038/nature 02593.
- Le Bagousse-Pinguet, Y., S. Soliveres, N. Gross, R. Torices, M. Berdugo, and F. T. Maestre. 2019. "Phylogenetic, Functional, and Taxonomic Richness Have both Positive and Negative Effects on Ecosystem Multifunctionality." *Proceedings of the National Academy of Sciences of the United States of America* 116: 8419–8424. https://doi.org/10.1073/pnas.181 5727116.
- Lehtinen, S., T. Tamminen, R. Ptacnik, and T. Andersen. 2017. "Phytoplankton Species Richness, Evenness, and Production in Relation to Nutrient Availability and Imbalance." *Limnology and Oceanography* 62: 1393–1408. https:// doi.org/10.1002/lno.10506.
- Leinster, T., and C. A. Cobbold. 2012. "Measuring Diversity: The Importance of Species Similarity." *Ecology* 93: 477–489. https://doi.org/10.1890/10-2402.1.
- Li, W. K. W. 2002. "Macroecological Patterns of Phytoplankton in the Northwestern North Atlantic Ocean." *Nature* 419: 154–157. https://doi.org/10.1038/nature00994.
- Liu, S. M., J. Zhang, H. T. Chen, and G. S. Zhang. 2005. "Factors Influencing Nutrient Dynamics in the Eutrophic Jiaozhou Bay, North China." *Progress in Oceanography* 66: 66– 85. https://doi.org/10.1016/j.pocean.2005.03.009.
- Liu, Z., H. Wei, G. Liu, and J. Zhang. 2004. "Simulation of Water Exchange in Jiaozhou Bay by Average Residence Time Approach." *Estuarine, Coastal and Shelf Science* 61: 25– 35. https://doi.org/10.1016/j.ecss.2004.04.009.
- Lukić, D., R. Limberger, S. Agatha, D. J. S. Montagnes, and T. Weisse. 2022. "Thermal Performance of Planktonic Ciliates Differs Between Marine and Freshwaters: A Case Study Providing Guidance for Climate Change Studies." *Limnology* and Oceanography Letters 7: 520–526. https://doi.org/10. 1002/lol2.10264.
- Mayfield, M. M., and J. M. Levine. 2010. "Opposing Effects of Competitive Exclusion on the Phylogenetic Structure of Communities." *Ecology Letters* 13: 1085–1093. https:// doi.org/10.1111/j.1461-0248.2010.01509.x.
- Mittelbach, G. G., C. F. Steiner, and S. M. Scheiner. 2001. "What Is the Observed Relationship Between Species Richness and Productivity?" *Ecology* 82: 2381–2396. https://doi. org/10.1890/0012-9658(2001)082[2381:WITORB]2.0.CO;2.

- Naeem, S., C. Prager, B. Weeks, et al. 2016. "Biodiversity as a Multidimensional Construct: A Review, Framework and Case Study of herbivory's Impact on Plant Biodiversity." *Proceedings of the Royal Society B* 283: 20153005. https://doi. org/10.1098/rspb.2015.3005.
- Pacala, S. W., and D. Tilman. 1994. "Limiting Similarity in Mechanistic and Spatial Models of Plant Competition in Heterogeneous Environments." *American Naturalist* 143: 222–257. https://doi.org/10.1086/285602.
- Prowe, A. E. F., M. Pahlow, and A. Oschlies. 2012. "Controls on the Diversity–Productivity Relationship in a Marine Ecosystem Model." *Ecological Modelling* 225: 167–176. https:// doi.org/10.1016/j.ecolmodel.2011.11.018.
- Santoferrara, L. F., and G. B., Mcmanus. 2017. "Integrating Dimensions of Biodiversity in Choreotrichs and Oligotrichs of Marine Plankton." *European Journal of Protistology* 61: 323–330. https://doi.org/10.1016/j.ejop.2017.04.004.
- Santoferrara, L. F., and G. B. Mcmanus. 2020. "Diversity and Biogeography as Revealed by Morphologies and DNA Sequences." In *Zooplankton Ecology*, edited by M. A. Teodósio and A. B. Barbosa, 85–118. Boca Raton: CRC Press.
- Scheiner, S. M., and S. Jones. 2002. "Diversity, Productivity and Scale in Wisconsin Vegetation." *Evolutionary Ecology Research* 4: 1097–1117.
- Smith, V. H. 2007. "Microbial Diversity–Productivity Relationships in Aquatic Ecosystems: Diversity–Productivity Relationships." *FEMS Microbiology Ecology* 62: 181–186. https:// doi.org/10.1111/j.1574-6941.2007.00381.x.
- Stirling, A. 2007. "A General Framework for Analysing Diversity in Science, Technology and Society." *Journal of the Royal Society Interface* 4: 707–719. https://doi.org/10.1098/ rsif.2007.0213.
- Stoecker, D., L. H. Davis, and A. Provan. 1983. "Growth of *Favella* Sp. (Ciliata: Tintinnina) and Other Microzooplankters in Cages Incubated In Situ and Comparison to Growth In Vitro." *Marine Biology* 75: 293–302. https:// doi.org/10.1007/BF00406015.
- Vadrucci, M. R., F. Vignes, A. Fiocca, et al. 2003. "Space–Time Patterns of co-Variation of Biodiversity and Primary Production in Phytoplankton Guilds of Coastal Marine Environments." *Aquatic Conservation: Marine and Freshwater Ecosystems* 13: 489–506. https://doi.org/10.1002/aqc.590.
- Vallina, S. M., M. J. Follows, S. Dutkiewicz, J. M. Montoya, P. Cermeno, and M. Loreau. 2014. "Global Relationship Between Phytoplankton Diversity and Productivity in the Ocean." *Nature Communications* 5: 4299. https://doi.org/10. 1038/ncomms5299.
- Verity, P. G. 1985. "Grazing, Respiration, Excretion, and Growth Rates of Tintinnids." *Limnology and Oceanography* 30: 1268–1282. https://doi.org/10.4319/lo.1985.30.6.1268.
- Verity, P. G. 1986. "Growth Rates of Natural Tintinnid Populations in Narragansett Bay." *Marine Ecology Progress Series* 29: 117–126. https://doi.org/10.3354/mep s029117.

- Verity, P. G., and C. Langdon. 1984. "Relationships Between Lorica Volume, Carbon, Nitrogen, and ATP Content of Tintinnids in Narragansett Bay." *Journal of Plankton Research* 6, no. 5: 859–868. https://doi.org/10.1093/plankt/6.5.859.
- Wang, C., C. Zhao, B. Zhou, et al. 2024. "Latitudinal Pronounced Variations in Tintinnid (Ciliophora) Community at Surface Waters From the South China Sea to the Yellow Sea: Oceanic-to-Neritic Species Shift, Biotic–Abiotic Interaction and Future Prediction." *Science of the Total Environment* 912: 169354. https://doi.org/10.1016/j.scitotenv.2023.169354.
- Wang, F., X. Li, and X. Tang. 2023. "The Seas Around China in a Warming Climate." *Nature Reviews Earth and Environment* 4, no. 8: 535–551. https://doi.org/10.1038/s43017 -023-00453-6.
- Weisse, T. 2024. "Physiological Mortality of Planktonic Ciliates: Estimates, Causes, and Consequences." *Limnology and Oceanography* 69: 524–532. https://doi.org/10.1002/lno. 12503.
- Whittaker, R. J. 2010. "Meta-Analyses and Mega-Mistakes: Calling Time on Meta-Analysis of the Species Richness– Productivity Relationship." *Ecology* 91: 2522–2533. https:// doi.org/10.1890/08-0968.1.
- Willig, M. R. 2011. "Biodiversity and Productivity." *Science* 333: 1709–1710. https://doi.org/10.1126/science.1212453.
- Witman, J. D., M. Cusson, P. Archambault, A. J. Pershing, and N. Mieszkowska. 2008. "The Relation Between Productivity and Species Diversity in Temperate-Arctic Marine Ecosystems." *Ecology* 89: S66–S80. https://doi.org/10.1890/07-1201.1.
- Worm, B., H. K. Lotze, H. Hillebrand, and U. Sommer. 2002. "Consumer Vs. Resource Control of Species Diversity and Ecosystem Functioning." *Nature* 417: 848–851. https://doi. org/10.1038/nature00830.
- Xu, H., B. Wang, M. Xin, et al. 2020. "Long-Term Changes of Nutrients and Their Ecological Effects in the Jiaozhou Bay (in Chinese With English Abstract)." *Advances in Marine Science* 38: 276–286. https://doi.org/10.3969/j.issn.1671-6647. 2020.02.008
- Zhang, W., M. Feng, Y. Yu, C. Zhang, and T. Xiao. 2012. *An Illustrated Guide to Contemporary Tintinnids in the World* (In Chinese). Beijing: Science Press.
- Zhao, Y., S. Zheng, X. Sun, and S. Sun. 2020. "A Dataset of Chlorophyll-*a* Concentration in the Jiaozhou Bay From 1997 to 2010 (In Chinese)." *CSdata* 5: 1–9. https://doi.org/ 10.11922/csdata.2019.0039.zh.

# **Supporting Information**

Additional Supporting Information may be found in the online version of this article.

Submitted 25 January 2024 Revised 16 January 2025 Accepted 10 February 2025